Author's response to reviews

Title: When is perigraft air pathological?

Authors:

Elizabeth Ball (liz_ball@yahoo.com)
Gareth Morris-Stiff (garethmorrisstiff@hotmail.com)
Mari Coxon (maricoxon@hotmail.com)
Michael H Lewis (mike.lewis@pr-tr.wales.nhs.uk)

Version: 2 Date: 6 April 2007

Author's response to reviews:

The only revision that was requested was to change the manuscript title to include 'a case report'.
I can also declare that there no competing interests in relation to this case report.
MH Lewis and G Morris-Stiff were responsible for the concept, E Ball and M Coxon wrote the paper and the
manuscript was reviewed and edited by G Morris-Stiff and MH Lewis. All authors approved the final version.

Patient consent was sought and received in relation to publication of this case report.